Author’s response to reviews

Title: Laparoscopic pyeloplasty of a completely duplicated collecting system.

Authors:

Konstantinos G Stravodimos (kgstravod@yahoo.com)
Ioannis Anastasiou (ianastasiou@med.uoa.gr)
Ioannis Adamakis (yanton@hotmail.com)
Theodoros Kapetanakis (thodoriskap@yahoo.com)
Georgios Koritsiadis (koritsiadisd@yahoo.gr)
Constantinos Constantinides (cconstan@med.uoa.gr)

Version: 4 Date: 11 August 2008

Author's response to reviews: see over
COVER LETTER

RESPONSES TO THE REVIEWERS’ COMMENTS

To the Editor-In-Chief,

To the Reviewers,

Dear Sirs,

Once more, we are grateful for your attention on our work.

We are pleased to see that you are satisfied with the revisions incorporated on the original draft according to your guidance. Concerning the comment of Dr Kumar, we do agree that the option of spatulating the lower ureter laterally and towards the lower pole of the kidney makes a viable choice for this kind of surgery. We therefore added a comment on this, stressing the aforementioned option in addition to what we actually did in our case (discussion, last sentence of the 4th paragraph).

We are anticipating for your final views, and remain sincerely yours,

Konstantinos G. Stravodimos MD, PhD. Assistant Professor of Urology, 1st Urology Department, University of Athens Medical School, Athens, Greece

Theodoros Kapetanakis MD. Resident in Urology, 1st Urology Department, University of Athens Medical School, Athens, Greece